\begin{table}%t1
\caption{\label{table_excluded}Transiting planetary systems excluded from the present analysis.}
   %\centering
\par
               \begin{tabular}{lccl}
               \hline\hline
               Name  & $e$ & $\lambda$ & References \\
                  \hline
               HD~17156 & $0.6819 \pm 0.0044$ & & \citet{Barbierietal09} \\
               HD~80606 & $0.9332 \pm 0.0008$ & & \citet{FisherValenti05,Pontetal09} \\
               HD~147506/HAT-P-2 & $0.5163 \pm 0.025$ & & \citet{Bakosetal07,Loeilletetal08} \\
               GJ~436 & $0.160 \pm 0.019$ & & \citet{Beanetal08,Demoryetal07,Manessetal07}\\
               XO-3 & $0.2884 \pm 0.0035$ & $37^{\circ} \pm 4^{\circ}$ & 
                               \citet{Johns-Krulletal08,Winnetal09} \\
               HAT-P-7 & & $182^{\circ} \pm 10^{\circ}$ & \citet{Naritaetal09,Winnetal09a} \\
               HAT-P-11 & $0.198 \pm 0.046$ & & \citet{Dittmannetal09,Bakosetal09b} \\
               WASP-14 & $0.091 \pm 0.003$ & $-33^{\circ} \pm 8^{\circ}$ & \citet{Joshietal09,Johnsonetal09a} \\
               WASP-17 & $0.129 \pm 0.1$ & $-147^{\circ} \pm 50^{\circ}$ & \citet{Andersonetal09} \\
               CoRoT-1 & & $77^{\circ} \pm 11^{\circ}$ & Pont et~al. (2010) \\
                    \hline
                           \end{tabular}\vspace*{3mm}
                             \end{table}